Author's response to reviews

Title: Do English Healthcare Settings Use 'Choice Architecture' Principles in Promoting Healthy Living for People with Psoriasis? An observational study

Authors:

Christopher Keyworth (chris.keyworth@manchester.ac.uk)
Pauline A Nelson (pauline.nelson@manchester.ac.uk)
Lis Cordingley (lis.cordingley@manchester.ac.uk)
Christopher EM Griffiths (christopher.griffiths@manchester.ac.uk)
Chris Bundy (christine.bundy@manchester.ac.uk)

Version: 2 Date: 12 January 2015

Author's response to reviews: see over
12th January 2015

Dear Ms. Dalumpines,

Re: Do English Healthcare Settings Use ‘Choice Architecture’ Principles in Promoting Healthy Living for People with Psoriasis? An observational study

Thank you for your recent correspondence regarding our above manuscript. Please find below a table outlining the specific changes made as requested.

<table>
<thead>
<tr>
<th>Comment</th>
<th>Changes made</th>
</tr>
</thead>
<tbody>
<tr>
<td>Line Numbering</td>
<td>Line numbering has been added to the main text file of the manuscript.</td>
</tr>
<tr>
<td>Please revise your manuscript to include line and page numbers.</td>
<td></td>
</tr>
<tr>
<td>Please update your ethics statement to include the name of the ethics committee that approved your study.</td>
<td>We have added the name of the ethics committee. (Page 6, line 121).</td>
</tr>
<tr>
<td>Please provide a List of abbreviations used (if any).</td>
<td>We have added the list to the manuscript (Page 17, lines 354-356).</td>
</tr>
<tr>
<td>Please place the funding information to Acknowledgement section.</td>
<td>The funding details have been to the acknowledgements section (Page 17, lines 369-372).</td>
</tr>
</tbody>
</table>

Please do not hesitate to contact me should you require anything further.

Yours sincerely

Mr Chris Keyworth on behalf of the study team
University of Manchester, UK.